Author's response to reviews

Title: Acute and chronic antihypertensive effects of Cinnamomum zeylanicum stem bark methanol extract in L-NAME-induced hypertensive rats

Authors:

Paulin Nyadjeu (epauka@yahoo.fr)
Elvine Pami Nguelefack-Mbuyo (mbuyopamielvine@gmail.com)
Albert Donatien Atsamo (atsamoalbert@yahoo.fr)
Telesphore Benoît Nguelefack (nguelefack@yahoo.fr)
Alain Bertrand Dongmo (alainberd@yahoo.fr)
Albert Kamanyi (Akamanyi@hotmail.com)

Version: 8 Date: 5 January 2013

Author's response to reviews: see over
 COVER LETTER FOR REVISED MANUSCRIPT  
Douala, the 05/01/2013  

MS: 1373889339731710

Subject: SUBMISSION OF THE REVISED MANUSCRIPT

Dear Editor,

I am enclosing herewith the revised manuscript entitled “Acute and chronic antihypertensive effects of Cinnamomum zeylanicum stem bark methanol extract in L-NAME-induced hypertensive rats.” for reevaluation and publication in Journal BMC.

We are waiting for the decision that will be reserved to it.

Sincerely yours.

ANSWER TO THE QUERIES

This work was carried out while the institutional guidelines for laboratory animal use and care were in creation. Thus, a temporally system of approval has been undertaken in our institution before any animal experiments can be undertaken. We received the approval of this commission before the beginning of our studies.

The modification has been added in red colour in the manuscript.

Page 5 paragraph entitled: Ethical consideration

Instead of: “Experimental protocols used in this study strictly conformed to the internationally accepted standard ethical guidelines .....”

Read: “Experimental protocols used in this study were approved by the Laboratory committee according to the standard ethical guidelines ...”